Author's response to reviews

Title: Endovascular coiling versus neurosurgical clipping in patients with unruptured intracranial aneurysm: a systematic review

Authors:

Jin Seup Hwang (hjs0409@neca.re.kr)
Min Kyung Hyun (mk3three@gmail.com)
Hyun Joo Lee (hilee@neca.re.kr)
Ji Eun Choi (jechoi@neca.re.kr)
Jong Hee Kim (goldpaper2@neca.re.kr)
Na Rae Lee (cherishnr@neca.re.kr)
Jin-Won Kwon (jinwon@neca.re.kr)
EnJu Lee (mk3three@neca.re.kr)

Version: 4 Date: 12 January 2012

Author's response to reviews: see over
you have given the same email address for all the co-authors of this study. Before we proceed please can we ask that you include separate email addresses for all authors and include these in the submission system.